Author's response to reviews

Title: Mycotic Aneurysm of the Posterior Tibial Artery - A rare complication of bacterial endocarditis: a case report

Authors:

Sandeep Patel (sandeep99@doctors.net.uk)
Nigel D'Souza (nige@doctrs.net.uk)
Shashank Gurjar (svgurjar@gmail.com)
Jim Hewes (jimhewes@hotmail.com)

Version: 7 Date: 26 August 2008

Author's response to reviews: see over
To Whom It May Concern:

We have attempted to address the main concern with regards to this paper – that of the evidence for this being a mycotic aneurysm. We have re-written the section describing the histological findings to include specific findings suggesting that there was inflammatory change present consistent with an infective process. We have also included a histological photo to be added to the finished paper when published. We have, as yet, still been unable to find any reliable data as to the incidence of mycotic aneurysms and so have not commented on this.

Yours truly,
Mr. Sandeep Patel
On behalf of other authors.